

Department of Medicine Faculty Papers

Department of Medicine

11-10-2022

Disparities in Mobile Phone Use Among Adults With Type 2 Diabetes Participating in Clinical Trials 2017-2021

Lyndsay A Nelson

Sammy Alfonsi

Lauren M Lestourgeon

Lindsay S Mayberry

Follow this and additional works at: https://jdc.jefferson.edu/medfp

Part of the Medicine and Health Sciences Commons
<u>Let us know how access to this document benefits you</u>

This Article is brought to you for free and open access by the Jefferson Digital Commons. The Jefferson Digital Commons is a service of Thomas Jefferson University's Center for Teaching and Learning (CTL). The Commons is a showcase for Jefferson books and journals, peer-reviewed scholarly publications, unique historical collections from the University archives, and teaching tools. The Jefferson Digital Commons allows researchers and interested readers anywhere in the world to learn about and keep up to date with Jefferson scholarship. This article has been accepted for inclusion in Department of Medicine Faculty Papers by an authorized administrator of the Jefferson Digital Commons. For more information, please contact: JeffersonDigitalCommons@jefferson.edu.



Brief Communications

Disparities in mobile phone use among adults with type 2 diabetes participating in clinical trials 2017–2021

Lyndsay A. Nelson (1^{,2}, Samuel P. Alfonsi III³, Lauren M. Lestourgeon^{1,2}, and Lindsay S. Mayberry^{1,2,4}

¹Department of Medicine, Vanderbilt University Medical Center, Nashville, Tennessee, USA, ²Center for Health Behavior and Health Education, Vanderbilt University Medical Center, Nashville, Tennessee, USA, ³Sidney Kimmel Medical College, Thomas Jefferson University, Philadelphia, Pennsylvania, USA and ⁴Department of Biomedical Informatics, Vanderbilt University Medical Center, Nashville, Tennessee, USA

Corresponding Author: Lindsay S. Mayberry, PhD, MS, Department of Medicine, Division of General Internal Medicine and Public Health, 2525 West End Ave. Suite 450, Nashville, TN 37203, USA; lindsay.mayberry@vumc.org

Received 27 May 2022; Revised 27 September 2022; Editorial Decision 19 October 2022; Accepted 25 October 2022

ABSTRACT

Despite smartphone ownership becoming ubiquitous, it is unclear whether and where disparities persist in experience using health apps. In 2 diverse samples of adults with type 2 diabetes collected 2017–2018 and 2020–2021, we examined adjusted disparities in smartphone ownership and health app use by age, gender, race, education, annual household income, health insurance status, health literacy, and hemoglobin A1c. In the earlier sample (N=422), 87% owned a smartphone and 49% of those had ever used a health app. Participants with lower income or limited health literacy had \geq 50% lower odds of owning a smartphone. Comparatively, in the later sample (N=330), almost all participants (98%) owned a smartphone and 70% of those had ever used a health app; however, disparities in health app use closely mirrored disparities in smartphone ownership from 2017 to 2018. Our findings suggest device ownership is necessary but insufficient for assuming people will use apps to support their health.

Key words: mobile phone, smartphone, applications, health disparities, type 2 diabetes, digital divide

LAY SUMMARY

Mobile health (mHealth) programs provide a convenient and effective approach to supporting health for people with chronic conditions like diabetes. We examined individual characteristics associated with smartphone ownership and health app use in 2 separate samples of adults with type 2 diabetes, collected a few years apart (2017–2018 and 2020–2021). In the earlier sample (N= 422), 87% owned a smartphone and among those, 49% had used a health app. People with low income and people with limited health literacy had 61% and 50% lower odds of owning a smartphone, respectively. Comparatively, in the later sample (N= 330), almost all participants (98%) owned a smartphone and 70% of those had used a health app. However, older people (\geq 60 years old), people with a high school degree or less education, and people with limited health literacy had 78%, 58%, and 66% lower odds (respectively) of having used a health app. Our findings support thoughtful consideration of how and to whom apps are presented in both research and clinical contexts to avoid worsening health disparities. Smartphone access is not the only prerequisite for use of mHealth tools.

© The Author(s) 2022. Published by Oxford University Press on behalf of the American Medical Informatics Association.

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial License (https://creativecommons.org/licenses/by-nc/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited. For commercial re-use, please contact journals.permissions@oup.com

INTRODUCTION

Smartphone applications (apps) designed to support type 2 diabetes (T2D) self-management have proliferated in recent years and continue to develop at a rapid pace.^{1–3} Although specific features and components vary widely, most apps share basic functions that include providing education, alerts and reminders, self-monitoring, and/or communication for self-management support.^{4,5} Evidence is accumulating that apps can improve self-care and glycemic management.^{6–8} Moreover, research has advanced on identifying the functions that contribute to an app's effectiveness.⁹ However, understanding who has access to and experience with using apps has implications for who can ultimately benefit.¹⁰

The term "digital divide" has been used to describe the gap in Internet and mobile phone access across different sociodemographic groups. Recent reporting suggests disparities in access that have persisted for decades are narrowing.^{11–13} For example, smartphone ownership among US adults is becoming more common across different economic, educational, and racial/ethnic backgrounds.¹⁴ From this closing of the digital divide, the conclusion often follows that more people are using health apps; however, it is necessary to critically examine usage by individuals' characteristics to understand whether and where disparities persist.

Previous research has examined sociodemographic factors associated with smartphone and app use, but there are notable limitations. First, the data in most studies were collected at least 5 years ago^{15–18}; with technology access rapidly evolving, recent data are necessary to understand current trends. Furthermore, while many studies have examined bivariate associations between individuals' characteristics and mobile phone use, fewer have adjusted for key confounders.^{15,17,19} Lastly, limited research has specifically examined smartphone and health app use among people with T2D, for whom racial/ethnic minorities and persons with lower socioeconomic status (SES) disproportionately experience worse health outcomes and stand to benefit considerably from this technology.²⁰

Objective

We examined predictors of smartphone ownership and health app use in 2 separate samples of adults with T2D. Data for one sample were collected between May 2017 and December 2018 and the other between April 2020 and October 2021. We examined predictors of mobile phone use, including sociodemographic and clinical characteristics in each sample separately.

MATERIALS AND METHODS

Study participants

We used participant-reported data collected as part of 2 randomized controlled trials (RCTs) evaluating effects of mobile phonedelivered interventions for diabetes self-management.^{21,22} Both interventions used only basic mobile phone technology (ie, phone calls and texts) and participants were required to own a mobile phone and be able to text after training with a research assistant. The 2017–2018 sample was recruited from community health centers and academic medical center primary care clinics in and around Nashville, TN. The 2020–2021 sample was recruited from the academic medical center primary care clinics only. For both studies, potential participants were identified with electronic health record (EHR) data, mailed a letter describing the study, and then patients who did not opt-out were contacted by phone calls from study staff to assess interest and complete eligibility screening. For both samples, eligible participants were English-speaking adults diagnosed with T2D, prescribed at least one daily T2D medication, and had a recent elevated hemoglobin A1c (HbA1c) value or the absence of an HbA1c value in the EHR. Interested and eligible participants completed consent, a survey, and HbA1c test. Surveys were completed in-person or by phone with a research assistant, online with an emailed link, or via a mailed paper copy, per participant preference. The Vanderbilt University IRB approved all study procedures for both RCTs.

Measures

All participants self-reported sociodemographics including age, gender, race, ethnicity, years of education, income, and insurance status. We assessed health literacy with the Brief Health Literacy Screen (BHLS)²³⁻²⁵ which asks: "How often do you have someone help you read hospital or clinic materials?" and "How often do you have problems learning about your medical condition because of difficulty understanding written information?" scored 1 = never to 5 = always, and "How confident are you filling out medical forms by yourself?" scored 1 = not at all to 5 = extremely. The first 2 items are reverse coded. In both the 2017-2018 and 2020-2021 sample, the BHLS had acceptable internal consistency ($\alpha = .71$ and $\alpha = .66$, respectively). HbA1c was collected via venipuncture or point-ofcare by patients' clinic or using a mail-in HbA1c kit provided and analyzed by CoreMedica Laboratories (Lee's Summit, MO). Participants were asked whether they owned a smartphone: ("Some cell phones are called 'smartphones' because they let you access the Internet and have a touchscreen. For example, an iPhone is a smartphone. Is your cell phone a smartphone?"). If participants responded ves, they were asked whether they had ever accessed a health app ("Have you ever used a health 'app'? It is a program that you can download on your phone that may help you track or monitor things such as your diet, exercise, or sleep."). Response options for both items included Yes, No, Don't know, and Refused.

Statistical analyses

All analyses were performed using SPSS Version 24. For either sample, we only analyzed data among participants who responded either yes or no to the smartphone question. Accordingly, 10 and 5 participants were excluded in the 2017-2018 and 2020-2021 sample, respectively. We calculated summary statistics using mean and standard deviation or percent. Then, for meaningful contrasts, we dichotomized characteristics to examine group differences in mobile phone use variables. For age, we compared participants who were <60 and \geq 60, based on prior literature examining disparities in technology adoption²⁶⁻²⁸ and to achieve a near equal distribution of participants across categories. For race, we compared participants who were Non-Hispanic Black and Non-Hispanic White. For income, we compared participants who had an annual household income of <\$35000 and \geq \$35000 based on prior research.¹⁴ For insurance, we compared participants who were underinsured (public insurance only or uninsured) with those who were privately insured. For health literacy, we compared participants reporting adequate (all BHLS items with response >3) and limited health literacy (any BHLS item with response \leq 3) based on prior research.^{29,30} For descriptive purposes, we compared the samples on characteristics of interest using Chi-square tests of independence for categorical variables and Mann-Whitney U tests for continuous variables. All subsequent analyses used categorical variables.

We used multivariate logistic regression models to examine adjusted associations between participants' characteristics and mobile phone use. We confirmed our data met assumptions for logistic regression. Participants reporting a race other than Non-Hispanic Black or Non-Hispanic White or reporting as Hispanic were included in multivariate models, but coefficients are not reported because there were too few participants in those groups to draw conclusions. For any variable with ≥ 10 missing values, we included an indicator for missingness in the multivariate models. We used this approach because missingness on these variables (income and HbA1c) was likely not at random, and this approach ensures all participants are included in contrasts of interest for which they have data.³¹

RESULTS

Participant characteristics for both samples are detailed in Table 1. In the 2017–2018 sample, average age was 56.3 ± 9.6 years, about half (55%) were female, and about 40% were Non-Hispanic Black. In the 2020–2021 sample, average age was 56.8 ± 11.0 years, about half (48%) were female, and about one-third (28%) were Non-Hispanic Black. Characteristics across samples were similar with

Table 1. Participant characteristics

respect to age, gender, and HbA1c; however, the 2017–2018 sample had more participants who were Non-Hispanic Black and who had limited health literacy and lower SES (ie, education, income, health insurance status) (Table 1).

2017-2018 sample

When recruiting the 2017–2018 sample, only 3% (37/1244) of patients screened reported not owning a phone with texting capability. Most enrolled participants (87%; 365/422) said they owned a smartphone, and among those, 49% (180/365) said they had used a health app. One participant said they did not know if they had used a health app and was excluded from the model predicting health app use. Adjusted logistic regression results for the associations between participant characteristics and mobile phone use are reported in Table 2. Participants who had <\$35K in household income or had limited health literacy, had 61% and 50% lower odds of owning a smartphone, respectively. Among those who owned a smartphone, participants with \leq 12 years of education or <\$35K in household income, had 49% and 42% lower odds of having used a health app, respectively.

Characteristic	2017-2018	2020-2021	P-value
	$(N = 422)^{a}$	$(N = 330)^{b}$	
Age, years	56.31 ± 9.55	56.79 ± 11.01	.48
<60	249 (59)	189 (57)	.66
≥ 60	173 (41)	141 (43)	
Gender ^d			
Female	231 (55)	155 (48)	.06
Male	191 (45)	171 (52)	
Race/Ethnicity			
Non-Hispanic White	198 (47)	204 (62)	<.001
Non-Hispanic Black	170 (40)	79 (24)	
Non-Hispanic Other race	25 (6)	22 (7)	
Hispanic	26 (7)	23 (7)	
Education, years	14.06 ± 3.10	15.31 ± 2.91	<.001
≤12	173 (41)	65 (20)	<.001
>12	243 (58)	259 (80)	
Annual household income, USD			
<\$35 000	233 (60)	58 (18)	<.001
≥\$35 000	154 (40)	261 (82)	
Health insurance			
Underinsured	207 (49)	64 (19)	<.001
Privately insured	212 (51)	257 (79)	
Health literacy (BHLS) ^e	13.04 ± 2.55	13.58 ± 1.89	.05
Limited	154 (36)	71 (22)	<.001
Adequate	268 (64)	259 (79)	
Hemoglobin A1c, %	8.63 ± 2.04	8.72 ± 1.69	.13
<8.5	214 (52)	161 (51)	.45
≥ 8.5	200 (48)	152 (49)	

^aIn the 2017–2018 sample, 3 participants did not provide race, 6 did not provide education, 35 did not provide income, 3 did not provide insurance, and 8 did not provide hemoglobin A1c data.

^bIn the 2020–2021 sample, 2 participants did not provide race, 2 participants did not provide gender, 6 participants did not provide education, 11 did not provide income, 9 did not provide insurance, and 17 did not provide hemoglobin A1c data.

^cMann–Whitney U tests used for continuous variables; Chi-square tests of independence used for categorical variables.

^dResponse options for gender included male, female, and prefer to self-describe. Two participants in the 2020–2021 sample indicated that they preferred to self-describe their gender; one wrote in their gender as non-binary and one wrote in their gender as Zim.

^ePossible scores for the composite BHLS score range from 3 to 15 with higher scores indicating higher health literacy; participants with a score \leq 3 on any item are categorized as having limited health literacy.

USD, United States Dollars; BHLS, Brief Health Literacy Screen.

	2017–2018		2020–2021 ^a	
	Owns smartphone $(N = 422)$ AOR (95% CI)	Uses health app $(N = 364)$ AOR $(95\%$ CI)	Uses health app (N = 321) AOR (95% CI)	
\geq 60 years of age	0.600 (0.318, 1.131)	0.737 (0.457, 1.189)	0.222 (0.120, 0.411)	
Gender (male)	0.925 (0.507, 1.685)	0.861 (0.548, 1.353)	0.795 (0.450, 1.406)	
Non-Hispanic Black	1.002 (0.527, 1.905)	0.711 (0.438, 1.154)	0.899 (0.451, 1.791)	
<12 years of education	0.562 (0.302, 1.045)	0.513 (0.319, 0.824)	0.424 (0.209, 0.858)	
<\$35K annual income	0.390 (0.175, 0.870)	0.578 (0.343, 0.976)	0.722 (0.316, 1.648)	
Underinsured	0.988 (0.511, 1.913)	0.777 (0.472, 1.279)	0.804 (0.394, 1.642)	
Limited health literacy	0.497(0.274, 0.900)	0.810 (0.508, 1.290)	0.343 (0.181, 0.650)	
Hemoglobin A1c \geq 8.5%	0.859 (0.466, 1.583)	0.731 (0.465, 1.153)	0.978 (0.542, 1.766)	

Table 2. Participant characteristics independently associated with mobile phone use

Bold text indicates *P*-value \leq .05.

AOR, adjusted odds ratio; CI, confidence interval.

^aWe did not examine correlates of smartphone ownership in the 2020–2021 sample because almost all participants (98%) reporting owning a smartphone.

2020-2021 sample

When recruiting the 2020–2021 sample, only 1% (7/519) of patients screened reported not owning a phone with texting capability. Almost all (98%; 323/330) enrolled participants owned a smartphone and among those, 70% (225/323) said they had used a health app. Two participants said they did not know if they had used a health app and were excluded from the model predicting health app use. Because almost all participants reported owning a smartphone, we did not examine correlates of smartphone ownership. Among those who owned a smartphone, participants who were ≥ 60 years of age, had ≤ 12 years of education, or had limited health literacy had 78%, 58%, and 66% lower odds (respectively) of having used a health app (Table 2).

DISCUSSION

Mobile health (mHealth) programs are ideally suited for improving diabetes management in that they provide convenient and timely support to people in their daily lives, where self-management occurs. Apps are one form of mHealth which, while quickly growing in availability, may not be accessible to all people with T2D. We compared smartphone ownership and health app use across 2 samples of adults with T2D collected several years apart. In the 2017–2018 sample, we found independent disparities in smartphone ownership by participants' income and health literacy. Among participants who did own a smartphone (87%), those with less education or lower income had lower adjusted odds of having used a health app. Comparatively, in the 2020–2021 sample, almost all participants (98%) owned a smartphone; however, disparities in health app use closely mirrored the disparities in smartphone ownership from 2017 to 2018 (Figure 1).

The ubiquity in smartphone ownership, both nationally and in our most recent sample of adults with T2D, is encouraging news for mHealth researchers. However, our findings are consistent with other recent studies that show emerging evidence of a new type of digital inequality based on experience with using digital tools.^{16,32} Despite owning a smartphone and having access to technology, individuals may be hesitant with using all its functions. Apps, in particular, require technical skills (eg, setting up an account, logging in, navigating the interface) which can be overwhelming for those who are less tech-savvy.³³ In addition, individuals from underserved populations have expressed concerns over the privacy and security risks with transmitting their personal health information via digital devices.^{16,34} Trust is another factor which may limit adoption of health apps considering historical mistrust of healthcare systems among minoritized groups.^{33,34} Taken together, these findings suggest that device ownership or device provision is necessary but insufficient for assuming people will use apps to support their health.^{35,36}

Several key design considerations can help ensure all individuals equally benefit from mHealth interventions. First, despite their capabilities, an app may not be the best choice for the intervention; rather, a phone could be leveraged for the functionality used by the most individuals.³⁷ Text messaging is ubiquitous across racial and socioeconomic groups and continues to be the most widely adopted and least expensive function on mobile phones.³⁸ In either RCT for the samples used in this study, less than 4% of all patients screened reported not owning a mobile phone with texting capability. Second, if an app is the necessary modality, user-centered and participatory design is key to ensure the tool is acceptable, appropriate, and meets end-users' needs.³⁴ Third, including technical support as part of the intervention is imperative.³⁴ As one example, Liu et al³⁹ improved mHealth adoption among low-income patients by having community health workers teach participants about the app at study enrollment. To provide this type of support more efficiently, a digital healthcare literacy screener could help identify those who require assistance, whether as part of a research study or in a healthcare setting.40

Our study has strengths and limitations with respect to generalizability. Data were collected from a single region in Middle Tennessee among adults with T2D and therefore our findings may not generalize to other regions and patient populations. While both samples were diverse, the 2020–2021 sample was more affluent than the 2017–2018 sample. However, the direction of this bias would have reduced our ability to detect disparities in the 2020–2021 sample and yet we identified similar disparities as in the 2017–2018 sample. Because participants in both samples signed up for an RCT examining diabetes self-management support delivered via mobile phones, they may not represent all adults with diabetes, but they do represent individuals interested in this type of support who would in theory be inclined to use health apps. We also acknowledge that dichotomizing our predictor variables may have reduced the power to detect associations between patient characteristics and mobile



Figure 1. Participant characteristics associated with smartphone ownership in the 2017–2018 sample and health app use in the 2020–2021 sample. An asterisk indicates *P*-value <.05.

phone use; however, the categories were selected based on existing literature identifying meaningful disparities in technology use. Finally, our study focused on examining differences in mobile phone use and did not assess reasons why participants were more or less likely to use health apps. More mixed-methods work in this area is necessary to inform mHealth deployment and design.

CONCLUSION

Given the increase in access to mobile technologies recently seen in low-income and minoritized racial and ethnic groups, mHealth has been championed as a strategy for improving population health and reducing health disparities. Our results and those of similar studies, support thoughtful consideration of how and to whom mHealth apps are presented. Without concerted efforts to address disparities in app use, there is potential to exacerbate existing disparities in health.

FUNDING

This research is funded by the National Institutes of Health NIH/ NIDDK (R01-DK100694 and R01-DK119282). The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health.

AUTHOR CONTRIBUTIONS

Authorship contributions according to the CRediT: LAN: Conceptualization, Formal Analysis, Writing—Original Draft, Writing— Review & Editing. SPA: Formal Analysis, Writing—Review & Editing. LML: Data Curation, Formal Analysis, Writing—Review & Editing. LSM: Conceptualization, Methodology, Investigation, Supervision, Funding Acquisition, Writing—Review & Editing.

ACKNOWLEDGMENTS

The authors thank our partnering clinics (ie, Faith Family Medical Center, The Clinic at Mercury Courts, Connectus Health, Shade Tree Clinic, Neighborhood Health, Vanderbilt Adult Primary Care), and the participants for their contributions to this research.

CONFLICT OF INTEREST STATEMENT

None declared.

DATA AVAILAVILITY

The data underlying this article will be shared on reasonable request to the corresponding author.

REFERENCES

- Doupis J, Festas G, Tsilivigos C, *et al.* Smartphone-based technology in diabetes management. *Diabetes Ther* 2020; 11 (3): 607–19.
- Lum E, Jimenez G, Huang Z, et al. Decision support and alerts of apps for self-management of blood glucose for type 2 diabetes. JAMA 2019; 321 (15): 1530–2.
- Massey A. Apps for Diabetes. Diabetes Self-Management. Secondary Apps for Diabetes. Diabetes Self-Management. 2021. https://www.diabetesselfmanagement.com/education/apps-for-diabetes/. Accessed April 15, 2022.
- Chomutare T, Fernandez-Luque L, Arsand E, *et al*. Features of mobile diabetes applications: review of the literature and analysis of current applica-

- Debon R, Coleone JD, Bellei EA, *et al.* Mobile health applications for chronic diseases: a systematic review of features for lifestyle improvement. *Diabetes Metab Syndr* 2019; 13 (4): 2507–12.
- Hou C, Carter B, Hewitt J, et al. Do mobile phone applications improve glycemic control (HbA1c) in the self-management of diabetes? A systematic review, meta-analysis, and grade of 14 randomized trials. *Diabetes Care* 2016; 39 (11): 2089–95.
- Wang Y, Min J, Khuri J, *et al.* Effectiveness of mobile health interventions on diabetes and obesity treatment and management: systematic review of systematic reviews. *JMIR Mhealth Uhealth* 2020; 8 (4): e15400.
- He Q, Zhao X, Wang Y, et al. Effectiveness of smartphone applicationbased self-management interventions in patients with type 2 diabetes: a systematic review and meta-analysis of randomized controlled trials. J Adv Nurs 2022; 78 (2): 348–62.
- Adu MD, Malabu UH, Callander EJ, *et al.* Considerations for the development of mobile phone apps to support diabetes self-management: systematic review. *JMIR Mhealth Uhealth* 2018; 6 (6): e10115.
- Huh J, Koola J, Contreras A, *et al.* Consumer health informatics adoption among underserved populations: thinking beyond the digital divide. *Yearb Med Inform* 2018; 27 (1): 146–55.
- Vangeepuram N, Mayer V, Fei K, *et al.* Smartphone ownership and perspectives on health apps among a vulnerable population in East Harlem, New York. *Mhealth* 2018; 4: 31.
- 12. Anderson-Lewis C, Darville G, Mercado RE, *et al.* Mhealth technology use and implications in historically underserved and minority populations in the United States: systematic literature review. *JMIR Mhealth Uhealth* 2018; 6 (6): e8383.
- Stockman MC, Modzelewski K, Steenkamp D. Mobile health and technology usage by patients in the diabetes, nutrition, and weight management clinic at an urban academic medical center. *Diabetes Technol Ther* 2019; 21 (7): 400–5.
- Pew Research Center. Mobile Fact Sheet. Secondary Mobile Fact Sheet. 2021. https://www.pewresearch.org/internet/fact-sheet/mobile/. Accessed September 14, 2022.
- Carroll JK, Moorhead A, Bond R, *et al.* Who uses mobile phone health apps and does use matter? A secondary data analytics approach. J Med Internet Res 2017; 19 (4): e5604.
- Bol N, Helberger N, Weert JCM. Differences in mobile health app use: a source of new digital inequalities? *Inf Soc* 2018; 34 (3): 183–93.
- Patel V, Barker W, Siminerio E. Disparities in Individuals' Access and Use of Health Information Technology in 2014. HealthIT. Washington, DC: Office of the National Coordinator for Health Information Technology, 2016.
- Mahmood A, Kedia S, Wyant DK, *et al.* Use of mobile health applications for health-promoting behavior among individuals with chronic medical conditions. *Digit Health* 2019; 5: 2055207619882181.
- Krebs P, Duncan DT. Health app use among US mobile phone owners: a national survey. JMIR mHealth uHealth 2015; 3 (4): e4924.
- 20. Ernsting C, Stühmann LM, Dombrowski SU, et al. Associations of health app use and perceived effectiveness in people with cardiovascular diseases and diabetes: population-based survey. *JMIR Mhealth Uhealth* 2019; 7 (3): e12179.
- Nelson LA, Wallston KA, Kripalani S, *et al.* Mobile phone support for diabetes self-care among diverse adults: protocol for a three-arm randomized controlled trial. *JMIR Res Protoc* 2018; 7 (4): e92.

- 22. Mayberry LS, El-Rifai M, Nelson LA, *et al.* Rationale, design, and recruitment outcomes for the family/friend activation to motivate self-care (FAMS) 2.0 randomized controlled trial among adults with type 2 diabetes and their support persons. *Contemp Clin Trials* 2022; 122: 106956.
- 23. Chew LD, Bradley KA, Boyko EJ. Brief questions to identify patients with inadequate health literacy. *Fam Med* 2004; 36 (8): 588–94.
- Wallston KA, Cawthon C, McNaughton CD, *et al.* Psychometric properties of the brief health literacy screen in clinical practice. *J Gen Intern Med* 2014; 29 (1): 119–26.
- Sarkar U, Schillinger D, Lopez A, *et al.* Validation of self-reported health literacy questions among diverse English and Spanish-speaking populations. J Gen Intern Med 2011; 26 (3): 265–71.
- Hargittai E, Piper AM, Morris MR. From internet access to internet skills: digital inequality among older adults. *Univ Access Inf Soc* 2019; 18 (4): 881–90.
- 27. Choudrie J, Pheeraphuttranghkoon S, Davari S. The digital divide and older adult population adoption, use and diffusion of mobile phones: a quantitative study. *Inf Syst Front* 2020; 22 (3): 673–95.
- Hoque R, Sorwar G. Understanding factors influencing the adoption of mhealth by the elderly: an extension of the Utaut model. *Int J Med Inform* 2017; 101: 75–84.
- Chew LD, Griffin JM, Partin MR, et al. Validation of screening questions for limited health literacy in a large VA outpatient population. J Gen Intern Med 2008; 23 (5): 561–6.
- Wallace LS, Rogers ES, Roskos SE, et al. Brief report: screening items to identify patients with limited health literacy skills. J Gen Intern Med 2006; 21 (8): 874–7.
- Zhuchkova S, Rotmistrov A. How to choose an approach to handling missing categorical data: (un) expected findings from a simulated statistical experiment. *Qual Quant* 2022; 56 (1): 1–22.
- 32. Adu MD, Malabu UH, Malau-Aduli AE, et al. Users' preferences and design recommendations to promote engagements with mobile apps for diabetes self-management: multi-national perspectives. PLoS One 2018; 13 (12): e0208942.
- Campos-Castillo C, Mayberry LS. Disparities in digital health in underserved populations. In: Klonoff DC, Kerr D, Weitzman ER, eds. *Diabetes Digital Health and Telehealth*. Elsevier Inc.; 2022: 269–80.
- Veinot TC, Mitchell H, Ancker JS. Good intentions are not enough: how informatics interventions can worsen inequality. J Am Med Inform Assoc 2018; 25 (8): 1080–8.
- Droge-Young L. Popular Mobile Health Apps Fail to Serve Vulnerable Populations. 2016. https://www.ucsf.edu/news/2016/07/403551/popularmobile-health-apps-fail-serve-vulnerable-populations. Accessed April 15, 2022.
- 36. Fox G, Connolly R. Mobile health technology adoption across generations: narrowing the digital divide. *Info Systems J* 2018; 28 (6): 995–1019.
- Mayberry LS, Jaser SS. Should there be an app for that? The case for text messaging in mHealth interventions. J Intern Med 2018; 283 (2): 212–3.
- Willcox JC, Dobson R, Whittaker R. Old-fashioned technology in the era of "bling": is there a future for text messaging in health care? *J Med Internet Res* 2019; 21 (12): e16630.
- Liu P, Astudillo K, Velez D, et al. Use of mobile health applications in low-income populations: a prospective study of facilitators and barriers. *Circ Cardiovasc Qual Outcomes* 2020; 13 (9): e007031.
- Nelson LA, Pennings JS, Sommer EC, et al. A 3-item measure of digital health care literacy: development and validation study. JMIR Form Res 2022; 6 (4): e36043.